\documentclass[11pt]{article}
\usepackage{fullpage,amsmath,amsthm}
\setlength{\parindent}{0in}
\setlength{\parskip}{2ex}

\begin{document}
\begin{minipage}[t]{0.5\textwidth}
\Large
CS 473 (Fall 2026)\\
\bf Problem X.Y
\end{minipage}
\hfill
\begin{minipage}[t]{0.5\textwidth}
\begin{flushright}
\Large
Name 1 (email address 1)\\
Name 2 (email address 2)\\
Name 3 (email address 3)
\end{flushright}
\end{minipage}

\hrulefill

\paragraph{Solution:}
Type your solutions here\ldots

Replace ``X.Y'' with the problem number, e.g., ``Problem 1.2''.
(If you have filled the form on your Gradescope identity, you may use an alias instead of your real name/university email address.)

Use a separate \LaTeX\ file for each problem.


\vfill

\paragraph{Collaborators:} (see the course policy and academic integrity web pages; write ``none'' if you haven't worked with anyone outside your group)

\paragraph{Sources:} (see the course policy and academic integrity web pages; write ``none'' if you haven't used any outside sources)

\paragraph{AI Use:} (see the course policy and academic integrity web pages; write ``none'' if you haven't used any generative AI tools)

\end{document}
